Author's response to reviews

Title: Supervised versus non-supervised implementation of an oral health care guideline in (residential) care homes: a cluster randomized controlled clinical trial.

Authors:

Gert-Jan van der Putten (gjvdputten@hetnet.nl)
Luc De Visschere (luc.devisschere@ugent.be)
Jos Schols (imschols@planet.nl)
Cees de Baat (c.debaat@dent.umcn.nl)
Jacques Vanobbergen (Jacques.Vanobbergen@Ugent.be)

Version: 4 Date: 2 July 2010

Author's response to reviews:

Dear Editor,

We added Table 1 in the documentfile, behind the reference list.

Yours sincerely,

Gert-Jan van der Putten